Author’s response to reviews

Title: Prevalence of obesity and abdominal obesity in the Lausanne population

Authors:

Pedro Marques-Vidal (pedro-manuel.marques-vidal@chuv.ch)
Murielle Bochud (Murielle.Bochud@chuv.ch)
Vincent Mooser (Vincent.2.Mooser@gsk.com)
Fred Paccaud (Fred.Paccaud@chuv.ch)
Gérard Waeber (Gerard.Waeber@chuv.ch)
Peter Vollenweider (Peter.Vollenweider@chuv.ch)

Version: 3 Date: 11 September 2008

Author’s response to reviews: see over
Concerne : article re-re-submission

Dear Sir:

Please find here enclosed the re-revised manuscript ref MS: 3515911391955194 untitled “Prevalence of obesity and abdominal obesity in the Lausanne population” by Murielle Bochud, Vincent Mooser, Fred Paccaud, Gérard Waeber, Peter Vollenweider and myself.

As requested, we included a statement indicating that the study was approved by the Ethic’s Committee of the University of Lausanne (see also Firmann et al, The CoLaus study: a population-based study to investigate the epidemiology and genetic determinants of cardiovascular risk factors and metabolic syndrome BMC Cardiovascular Disorders 2008, 8:6 doi:10.1186/1471-2261-8-6). We apologise for not having stated it immediately.

Hoping that the re-revised manuscript will be acceptable for publication, we thank you in advance for re-re-considering it.

With my best regards,

Pedro Marques-Vidal